Author's response to reviews

Title: Drinking behaviours and blood alcohol concentration in four European drinking environments: a cross-sectional study

Authors:

Karen Hughes (k.e.hughes@ljmu.ac.uk)
Zara Quigg (z.a.quigg@ljmu.ac.uk)
Mark A Bellis (m.a.bellis@ljmu.ac.uk)
Ninette van Hasselt (nhasselt@trimbos.nl)
Amador Calafat (irefrea@irefrea.org)
Matej Kosir (info@institut-utrip.si)
Montse Juan (irefrea@cop.es)
Mariangels Duch (mduch@irefrea.org)
Lotte Voorham (lvoorham@trimbos.nl)

Version: 3 Date: 17 September 2011

Author's response to reviews:

Dear Jimmar Dizon,

Please find attached a revised manuscript containing the following competing interest statement:

"In the past five years, Centre for Public Health received a grant from Drinkaware to undertake an independent study of drinking behaviours among students and MAB has provided them with independent medical advice. Drinkaware is an independent UK-wide charity supported by voluntary contributions from the alcohol and supermarket industries."

If you require any further revisions please do not hesitate to contact us.

Best wishes

Karen Hughes